Author’s response to reviews

Title: Menstrual Characteristics, Disorders And Associated Risk Factors Among Female International Students In Zhejiang Province, China: A cross-sectional survey.

Authors:

Emmanuel Ansong (dr.ansong1@gmail.com)
Samuel Kofi Arhin (samkoarhin@gmail.com)
Yaoyao Cai (359461572@qq.com)
Xinxin Xu (736114766@qq.com)
Xueqing Wu (wuxueqing.37@hotmail.com)

Version: 4 Date: 11 Dec 2018

Author’s response to reviews:

The Editor-In-Chief,
BMC, Women’s Health.

Dear Sir/Madam,

With reference to comments and suggestions concerning our article (ID: BMWH-D-17-00227R3): Menstrual Characteristics, Disorders and Associated Risk Factors among Female International Students in Zhejiang Province, China. We would like to express our utmost gratitude to reviewers for taking time and effort to read and give helpful suggestions to help improve our manuscript. We have therefore addressed major concerns below.

Answers to editor’s comment:

1. We have updated the manuscript title in the submission system.
2. We have also changed the ‘Materials and Method to ‘Methods’.
3. The figure title has also been removed from the figure file.
4. Concerning additional files, we do not intend to publish the supplementary sample questionnaire so we have subsequently removed it from the manuscript.

Answers to reviewer’s comments:

1. Results: We intend to report the median (± SD) for the BMI for the population. We have corrected the typo and the correction has been highlighted in the manuscript.
2. References: We have also adjusted the font size of the reference to the be exactly as that of the manuscript.
Please address all correspondence to:

Prof. Xueqing Wu

Department of Obstetrics and Gynaecology

The First Affiliated Hospital of Wenzhou Medical University

Nanbaixiang Street, Ouhai District.

Wenzhou Zhejiang 325000 .P.R.China .

E-mail: wuxueqing.37@hotmail.com

Tel: +86-13706661290

Fax: +86-577-80671137

Yours sincerely,

Prof. Xueqing Wu